Author's response to reviews

Title: Stem cell abundant protein Nanog, Nucleostemin and Musashi1 highly expressed in malignant cervical epithelial cells

Authors:

Feng Ye (yefyef@zju.edu.cn)
Caiyun Zhou (Caiyunzhou@163.com)
Qi Cheng (qicheng@sina.com)
Jiajie Shen (shenjiajie@yahoo.com.cn)
Huaizeng Chen (chenhz@zju.edu.cn)

Version: 3 Date: 20 March 2008

Author's response to reviews: see over
Dear Editor:

Our study was approved by the Medical Ethical Committee of Women’s Hospital, School of Medicine, Zhejiang University. All patients signed informed consent to allow molecular research on specimen obtained during surgical operation. We have clarified the consent/ethics declaration in the methods section of the study (Page 4, Paragraph 2). If you have any questions, please contact with me anytime.

Sincerely yours,

Best wish to you.

Huaizeng Chen, M D
Women's Reproductive Health Laboratory of Zhejiang Province
Women’s Hospital
School of Medicine
Zhejiang University
Hangzhou 310006, China
Phone:  86-571-87061501
Fax number: 86-571-87061878
E-mail: chenhz@zju.edu.cn